Author's response to reviews

Title: The World Health Organization Quality of Life Instrument for People with Intellectual and Physical Disabilities (WHOQOL-Dis): evidence of validity of the Brazilian version.

Authors:

    JULIANA BREDEMEIER (juliana.bredemeier@gmail.com)
    GABRIELA P WAGNER (gabrielapwagner@gmail.com)
    MARILYN AGRANONIK (marilyn.agranonik@gmail.com)
    TATIANA S PEREZ (tspalper@gmail.com)
    MARCELO P A FLECK (mpafleck@gmail.com)

Version: 2 Date: 23 July 2013

Author's response to reviews:

Dear editors,

We have followed your advice and have submitted this manuscript to a new review with a British English native speaker. We thank you very much on this opportunity.

Sincerely,

The Authors.